



# **Current Knowledge and Pending Research on Sulfate Resistance of Recycled Aggregate Concrete**

Lautaro R. Santillán <sup>1,2,\*</sup>, Claudio J. Zega <sup>2,3</sup> and Edgardo F. Irassar <sup>4,5,\*</sup>

- <sup>1</sup> Instituto de Investigaciones en Tecnología y Ciencias de la Ingeniería (IITCI CONICET-UNCo), Neuquén 8300, Argentina
- <sup>2</sup> Consejo Nacional de Investigaciones Científicas y Técnicas (CONICET) CCT-La Plata, La Plata 1900, Argentina; cj.zega@conivet.gov.ar
- <sup>3</sup> Laboratorio de Entrenamiento Multidisciplinario para la Investigación Tecnológica (LEMIT), Comisión de Investigaciones Científicas (CIC), La Plata 1900, Argentina
- <sup>4</sup> Departamento de Ingeniería Civil y Agrimensura, Universidad del Centro de la Provincia de Buenos Aires (UNCPBA), Olavarría 7400, Argentina
- <sup>5</sup> Centro de Investigaciones en Física e Ingeniería del Centro de la Provincia de Buenos Aires (CIFICEN), Olavarría 7400, Argentina
- \* Correspondence: lrsantillan@conicet.gov.ar (L.R.S.); firassar@fio.unicen.edu.ar (E.F.I.)

**Abstract:** The building sector's sustainability requires construction and demolition waste (CDW) to contribute to the circular economy. Among the CDW, recycled concrete aggregates (RA) have been mainly studied to replace natural aggregates. Still, the approval of their use in regulations and standards is slower. Some barriers to the adoption of RA are related to the durability of recycled aggregate concrete (RAC). However, their physical and mechanical properties have been extensively studied. The durability risks associated with sulfate attacks have been solved for conventional concrete. However, sulfate attack on recycled concrete still raises numerous unsolved questions. In this literature review, the experience of sulfate attack on RAC is compiled and analyzed using a compressive framework highlighting the most relevant aspects of the new matrix in RAC and the old matrix of RA to support its relevance to the damaging sulfate process. Suggestions for further research are presented to understand the full extent of this issue and contribute to incorporating and extending recycled aggregates into existing regulations.

Keywords: recycled concrete; recycled concrete aggregates; sulfate attack; review

#### 1. Introduction

The construction industry significantly impacts the environment due to its high material and energy consumption. Concrete has a significant environmental footprint because of the raw materials, energy, and CO<sub>2</sub> emissions associated with its production (about 5–7% of global emissions) [1,2]. However, one strategy that can improve concrete's sustainability is the use of CDW as aggregates in its production [2–4]. This approach offers several environmental benefits, including reducing the exploitation of natural resources, decreasing industry and municipal wastes for disposal, and significantly decreasing transportation costs [2,5,6]. Specifically, aggregates derived from concrete demolition, known as recycled concrete aggregates (RA), have been found to possess competitive characteristics for use in the construction industry [7,8].

Today, some building codes allow certain replacement percentages of recycled fine and coarse aggregates in structural and nonstructural concrete production [9,10]. In general, they set limits for the replacement of NA by RA according to the purpose of the concrete (strength class, environmental exposure class) and the source/type of RA (classification) that determines their properties [7,9–11]. Increasing the allowable replacement levels and the wide use of RA in different mixes will largely depend on a thorough understanding of RA's effects on the performance of the new concrete [12,13].



Citation: Santillán, L.R.; Zega, C.J.; Irassar, E.F. Current Knowledge and Pending Research on Sulfate Resistance of Recycled Aggregate Concrete. *Sustainability* **2024**, *16*, 1310. https://doi.org/10.3390/su16031310

Academic Editor: Constantin Chalioris

Received: 30 November 2023 Revised: 10 January 2024 Accepted: 30 January 2024 Published: 4 February 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The mechanical and physical properties of RAC and (and RAM) have been extensively studied. Generally, mechanical strength and stiffness are lower for RAC than for NAC due to its higher porosity [14–17]. RAC's transport properties are higher also due to the higher porosity [18–20]. Regarding their durability, although many studies have been carried out [21–25], the wide variety of attacks and methods used to evaluate them requires an exhaustive analysis of experimental campaigns for a complete understanding.

Sulfate attack (in all forms) is a well-known problem in concrete technology [26–32]. Proven technological solutions already exist for producing conventional sulfate-resistant concretes [33–35]. However, RACs (and RAMs) have very different properties from NACs, mainly higher porosity [14,18]. In addition, RAC presents a larger ITZ, which includes the interface between the old cement paste and the new cement matrix [36]. This issue requires a focused study of the sulfate problems for this new material.

For clarity, this paper is organized as follows: Section 2 presents the current state of knowledge on two issues related to the article's subject—sulfate attack and related durability problems and the properties of recycled aggregates and their effects on cementbased materials. Section 3 presents a comprehensive analysis of experimental results, classified by type of sulfate attack (ESA, PSA, and ISA) and, in the case of ESA, by material scale (RAC and RAM). Section 4 discusses the previous two chapters, focusing on the effect of RA, technological solutions to improve the quality of RAC, and recommended guidelines for further research. Finally, Section 5 presents the conclusions of this work.

### 2. Current Knowledge about Sulfate Attack: The Particular Case of Recycled Aggregate Concrete (RAC)

In the literature, many papers on RAC's durability focus on investigating the transport properties, as they are closely related to most durability issues [14,37–44]. In addition, several papers examine the performance of RAC under specific exposure conditions. Papers on sulfate attacks remain low [45–94], especially considering the large number of parameters that influence attack development. This section aims to provide an overview of the current knowledge of sulfate-related durability problems and highlight the key factors to consider when evaluating this attack on cement-based materials with RA.

#### 2.1. Sulfate-Related Attacks

Sulfate attack is a chemical attack with various forms, usually associated with other types of attack; it is extensively described by the literature and has several technological solutions for conventional concrete [26,32,33,95,96]. However, research on the mechanisms associated with the attack is still ongoing due to its complexity, the number of involved parameters, and the variety of measurement methods [97–99].

#### 2.1.1. Nomenclature

For a long time, the term "sulfate attack" was used to refer to durability problems caused by the sulfate ions in concrete. However, it is possible to distinguish different processes, and the nomenclature proposed by Neville [97] is adopted in this review. External sulfate attack (ESA) is the chemical attack that occurs on hydrated Portland cement paste in the presence of external sulfate ions that ingress into concrete by a transport mechanism. When the concrete components provide the sulfate ions, it is called an internal sulfate attack (ISA). Finally, real field structures exposed to a rich sulfate environment (soil or water) are subjected to moisture gradients, causing physical attack by salt crystallization. When sulfate attack" usually refers to ESA and PSA, as both phenomena are considered to be durability issues related to an aggressive environment. In contrast, ISA is related to material impurities, mixture design, or manufacturing process issues. The phenomenon known as delayed ettringite formation (DEF) is merely a specific case of ISA.

#### 2.1.2. Mechanisms

External sulfate attack (ESA) is a chemical reaction between some phases of hydrated Portland cement and sulfate ions from the environment. Sulfate ions can react with calcium hydroxide to form gypsum (Equation (1)), while the aluminate phases of cement combine with gypsum and additional sulfate ions to form ettringite (Equation (2)). These reaction products, particularly ettringite, occupy more space than the constituents from which they are derived and are responsible for the physical effects observed in the material [32]. These effects include swelling, cracking, and spalling due to the internal tensile stresses of ettringite precipitation. Swelling occurs due to the crystallization pressure of ettringite formation in a sulfate-supersaturated pore solution, not when all the pore volume of the concrete is filled with ESA products [100]. In later stages, portlandite consumption for gypsum formation leads to instability in the C-S-H, resulting in the softening of the material [26].

$$Ca^{2+} + SO_4^{2-} + 2H_2O \rightarrow CaSO_4 \cdot 2H_2O$$
<sup>(1)</sup>

$$6Ca^{2+} + 3SO_4^{2-} + 2Al(OH)_2^{-} + 8OH + 26H_2O \rightarrow 3CaO \cdot Al_2O_3 \cdot 3CaSO_4 \cdot 32H_2O$$
(2)

When the cementitious matrix has a significant carbonate content (e.g., due to the use of limestone filler), a further stage of attack develops in which the precipitating mineral is thaumasite. In general, thaumasite formation has been observed under low-temperature conditions, and the damage associated with this condition is more significant than that at ambient temperatures [101,102]. In magnesium sulfate attack,  $Mg^{2+}$  competes with  $SO_4^{2-}$  to combine with portlandite, resulting in brucite precipitation on the concrete surface. This poorly soluble and dense compound acts as a barrier, controlling the ingress of sulfate ions into the material. However,  $Mg^{2+}$  ions can also react with C-S-H, replacing the calcium ion and forming M-S-H, a nonbinding gel that produces the softening [103,104].

For moisture gradients in the material, the conditions for PSA are present. This physical attack is merely a particular case of salt crystallization. The damage mechanism consists of the precipitation of sulfate salts in the internal pores of the concrete after these salts have entered from the outside by different transport mechanisms (convection by capillary absorption or pressure gradient, diffusion). The crystallization pressure associated with the precipitation of these salts in the pores of the concrete causes internal cracking of the concrete by tensile stresses, increasing the transport of the aggressive solution in the material for further advancement of the attack [30,105,106].

Finally, the chemical mechanism of ISA is similar to that of ESA, but in this case, the sulfates do not come from the environment but from the concrete components. For ISA damage, internal sulfates and sufficient moisture content must trigger the damaging reactions [107].

Two key factors that determine the performance of concrete against sulfate attack are its transport properties (kinetic condition) and mineralogical composition (thermodynamic condition). A more compact and dense material with lower transport properties is more resistant to penetration by aggressive ions, regardless of the mode of ion transport [108]. In the case of internal sulfate attack (ISA), where ions are already present in the material, a dense concrete would also delay the ingress of additional water, which is necessary to initiate the mechanism [109]. Moreover, the mineralogical composition of the concrete, specifically the aluminate phases and portlandite content in its cement paste, may affect its susceptibility to sulfate attack because these phases can react with sulfate to form harmful products [26,27,110]. In the case of ISA, the sulfate content of the concrete components, including cement and aggregates, is a crucial factor in the development of the attack.

#### 2.1.3. Evaluation Methods

Sulfate performance is usually evaluated by exposing specimens to different conditions and monitoring some characteristic properties over time. Several properties have been used to assess sulfate attack. The most common ones are expansion, mass loss, and drops in mechanical properties (strength and stiffness) [95,111–116], each having advantages and

disadvantages but good statistical results. Additionally, instrumental techniques such as XRD, XRF, thermogravimetry, FTIR, and SEM/EDS make it possible to characterize and quantify the profile of attack progress in terms of elemental oxides (i.e., SO<sub>3</sub> concentration) and mineralogical compounds (i.e., phase changes) at different scales (mesostructural and microstructural). There are even works that use electrochemical techniques [49,90], changes in microhardness at the interface [117], or fracture energy [55] to evaluate the progress of sulfate attack.

There is a large amount of variability in mixture performance depending on the test or exposure conditions. Effects such as the temperature, the specimen/solution volume ratio, the shape and size of the specimen, and the cation associated with sulfate, among others, affect the failure mechanisms [96,115]. Two fundamental setup variables are the sulfate concentration and the humidity regimen. The former defines the rate of diffusive transport (concentration gradient), and the latter activates or deactivates the convective transport of the solution (by capillary absorption or a pressure gradient). When specimens are saturated (i.e., immersed in sulfate solution), only ESA occurs [98,115,118,119]. In contrast, when specimens are subjected to wetting and drying cycles or partial saturation, the conditions for combined ESA and PSA occur [106,117,120,121]. The exposure conditions used in the laboratory are generally more severe than those found in the field, as the aim is to accelerate the attack to obtain information within the timescales of an investigation.

#### 2.1.4. Technological Solutions

Technological solutions for sulfate-resistant concrete require physical and chemical improvements to the cement matrix and the ITZ. Physically, this is achieved by reducing the connected capillary porosity of the cement paste; chemically, improvements are made by decreasing the content of reactive phases in the cement. These solutions can be achieved by reducing the w/c ratio (use of chemical admixtures), using cement with low C<sub>3</sub>A content, and using active SCM (reduction in clinker factor, pozzolanic reaction). Current regulations have focused on these two solutions to develop sulfate-resistant concretes.

## 2.2. *Characteristics of Recycled Concrete Affecting Performance against Sulfate Attack* 2.2.1. RAC Aspects in Relation to Sulfate Attack

Two fundamental aspects of RAC are linked to sulfate attack on RAC: on the one hand, its higher porosity and higher interface content (kinetic aspect), and on the other hand, the high cement paste volume or high sulfate-reactive phases (thermodynamic aspect).

Generally, RAC is more porous than NAC due to the higher porosity of RA [18,22,42,122–126], the higher number of interfaces generated by RA [36,127–130], and the higher amount of entrapped air due to the shape of particles of RA [86,129]. For this reason, the strength and modulus of elasticity of a mixture with RA are often lower than those of the same mixture with NA. Numerous papers have reported higher transport properties of RAC than NAC [14,22,37,42,44,131], which is attributable to its high porosity.

Increased porosity would harm sulfate attack, allowing the ingress of more aggressive ions in concrete subjected to ESA or ESA + AFS or favoring the ingress of the water required for ISA development. However, increased porosity could also have some positive effects. Some recycled concretes performed better under specific proportions and exposure conditions than those with natural aggregates [58,59,62,65,67,81,91,94], suggesting that other positive effects of RA may compensate for the expected adverse effects.

Regarding chemical-mineral properties, the main difference between RAC and NAC regarding sulfate attack is the higher relative cement-paste content due to mortar or cement paste in the RA. Cement paste could contain sulfate-reactive phases (CH, AFm), leading to a more extensive attack. However, the content of additional reactive material would be limited by the ratio of replacement with RA and the attached mortar or cement-paste content of the RA. As an example, considering concrete containing 40% coarse aggregate, 40% fine aggregate, and 20% cement paste by weight, if 100% of the coarse aggregate is replaced by coarse RA containing 20% cement paste by weight (the expected value [132]),

the relative cement-paste content in the new concrete would increase to 28%. If, in addition, 100% of the fine aggregate is replaced by fine RA containing 30% cement paste (expected value [133]), the relative cement-paste content in the new concrete would increase to 40%. The potential reactivity of RA with external sulfate ions has yet to be thoroughly investigated and is one of the gaps in the current literature.

#### 2.2.2. RA Properties

The performance of RAC would be affected not only by the replacement ratio of RA but also by their physical and chemical properties. The relevant physical properties of RA are mortar/paste content, porosity, and absorption. These properties are related to the source concrete's properties and the crushing method [23,40,133]. The chemical properties of RA depend on the concrete source, especially the binder composition (cement and supplementary cementitious materials) and its use (sulfate contamination). In addition, the degree of carbonation of the cement paste in RA can also affect the performance of the aggregates when exposed to sulfate environments [22,123]. The content of attached mortar or cement paste has been suggested in the literature as a quality control parameter [134,135]. However, there is no reliable and unified method for this, and no methods have been proposed in the literature to assess the potential reactivity of the phases included within the RA.

Regarding ISA, the key factor for RA is their likely contamination with compounds necessary for the damaging reaction. In particular, the contamination of aggregates with sulfate-based constituents is a recurring problem since, depending on the extraction and separation process quality during the demolition phases, RA may have a high content of gypsum, a material widely used in construction. This problem, addressed by several papers in the literature [45,46,48,51,73,81], raises the question of the permissible limits for sulfate content in aggregates, which are usually limited by existing regulations.

Finally, the moisture content of RA is also an important aspect considered in several studies [15,23,40,136,137]. The main concern is about the higher porosity of RA with respect to NA and its consequent higher water absorption capacity. Depending on the moisture content of the aggregates at the time of mixing, they can change the effective w/c ratio of the mixture, especially at the interfaces, and thus significantly affect the transport properties of RAC. In addition, they can modify the curing process using the internal curing of the concrete [15,48,138,139].

Given the above, the adoption of criteria for the characterization and qualification of recycled aggregates appears to be a possible need, not only for their effect on the general properties of new concrete but also for their effect in the specific case of sulfate attack.

#### 2.2.3. Regulations

Several countries have already included the use of recycled aggregates in their regulations. Most of them establish a quality classification of recycled aggregates according to their composition or parameters, such as absorption and density, and then define the limits of the permissible contents and the purposes for which they can be used (strength level, structural use, type of exposure) [9,11,140]. The regulations generally allow the use of RA in nonstructural or structural concretes exposed to conditions of low aggressiveness to their durability.

Regulations also set requirements for concrete aggregates in general. Among these requirements, most regulations limit the content of contaminants that may affect the durability of concrete, for example, by setting the maximum content of soluble sulfates to values between 0.2% and 1.0% [9,83]. These limits can be restrictive in the case of RA, which are highly susceptible to contamination, depending on the process used to produce the RA or the environment to which the original concrete was exposed. Modifying the regulations to increase the allowable content of RACs increases the possible uses of RACs and the limited  $SO_3$  content, which will depend mainly on the research progress on this issue.

#### 3. Experimental Results Available in the Literature

As mentioned before, the number of research papers on the performance of recycled concrete against sulfate attack still needs to be increased in order to understand the degree of influence of each intervening variable. Although there are several publications on this topic, they vary significantly regarding research objectives, setup parameters, and evaluation methods. Many of these studies focus on characterizing specific types of recycled concrete aggregates (RA) or evaluating eco-efficient concretes that incorporate RA together with SCMs. These studies examine various properties, including sulfate resistance, but cover only a few variables. Some research also considers sulfate attack in combination with other forms of deterioration, such as freezing and thawing. Each paper uses different test conditions and evaluation methods, making comparing results and drawing meaningful conclusions across different studies challenging.

The literature data was first classified based on the predominant type of sulfate attack: ESA, PSA, or ISA. In order to compare the results of studies with different experimental setups and evaluation methods, a numerical factor was defined: the degradation ratio, which is the relation between the degradation observed for each evaluated RAC (or RAM) and its corresponding reference concrete (or mortar) in a given time. One example is the final expansion of RAC divided by the final expansion of the reference NAC (same criteria for compressive strength loss, weight variation, SO<sub>3</sub> incoming, et cetera). Degradation ratios greater than 1 indicate worse performance of RAC compared to NAC, while ratios below 1 indicate better performance.

The variety of existing test parameters explains the large variability in results in the literature. For example, no trend was found when correlating the deterioration ratio with the w/c ratio of concrete. This is because there are statistically very few results, and they come from tests with very different setups (exposure, evaluation method), making the results not directly comparable. This paper presents the degradation ratio in relation to the RA content, which is a commonly used parameter in literature reviews [21–25] and is significant for regulatory discussions. Regarding the ISA, only a few papers are available, and the experimental setups are quite diverse, making them difficult to compare. The key experimental setups and outputs are summarized and described in the text to conduct a qualitative analysis.

#### 3.1. ESA

#### 3.1.1. Recycled Aggregate Concrete (RAC)

Few studies [52,54,56,61,66,67,70–72,75,77,79,89,91] evaluate the performance of RAC against ESA exposure using immersed specimens. In addition, several works [55,56,61,74,77,78] investigate its performance against ESA in conjunction with other durability tests, thus limiting the campaigns and the number of parameters each test covers. Figure 1 shows the processed results of studies that have examined ESA in concrete with CRA.

In general, an increase in CRA content is correlated with an increase in the deterioration ratio. However, the few existing results of partial substitutions do not show a conclusive relationship between CRA content and performance [52,67,75,79]. For 100% CRA replacement, deterioration ratios ranged from 0.62 to 2.25. All cases with a degradation ratio lower than 1 were cases where 100% RA was combined with some SCMs [66,71,72,91]. Then, depending on other parameters such as matrix quality, the degradation ratio of concretes with 100% RA concretes varied between 1.06 and 1.73 (except for Bulatovic et al. [93], with a degradation ratio of 2.25).

The higher porosity of RA explains the increase in RAC deterioration compared to NAC deterioration. In this type of attack, the ingress of sulfate ions into the specimens is due to diffusion, and the rate of diffusion of sulfate ions increases as the total porosity increases [32]. No consistent relationship between RA content and deterioration rate is observed for concretes with partial replacement. Hence, Arafa et al. [79] and Santillán et al. [67] observed a worse sulfate performance for partial RAC replacements (60 and 75%,



**Figure 1.** Degradation ratios of concretes with different % CRA exposed to ESA [52,54,56,61,66,67,70–72,75,79,85,91,93].

The results of each experimental campaign show a substantial effect on the quality of the RAC cement matrix. For example, Bulatovic et al. [93] evaluated RAC with different cement types and w/c ratios (good- and poor-quality matrices). The RAC degradation ratio (100% RA) with a poor-quality matrix was 2.25 and 1.16 for sodium and magnesium sulfate exposure, respectively. In contrast, the RAC with good-quality matrices performed similarly to the reference concretes. According to Dhir et al. [52], concretes with a low strength level (w/c between 0.81 and 0.76) showed comparable performance to the reference mixtures for CRA replacement values up to 30%, while for higher replacement values, the deterioration increased, reaching a degradation ratio of 1.68 for 100% CRA replacement.

Some of the quoted papers specifically study the effect of SCM on the performance of recycled concretes. This is the case for Somna et al. [71], Tangchirapat et al. [72], Boudali et al. [91], and Rattanachu et al. [66]. In all of those studies, RACs were observed to have slightly worse performance than NACs (degradation ratio between 1.15 and 1.4), but performance was even better than that of the control when RAC had pozzolanic SCM (degradation ratio < 0.7). The positive effect of pozzolanic SCMs on the performance of concrete against sulfates is well known [35,49,111,141]. The secondary reaction of SCMs consumes the portlandite required for ESA. In addition, it refines the pores by forming C-S-H, significantly reducing the concrete's transport properties. In the case of RAC, equal or even higher efficiency is observed for SCM compared to conventional concrete [65,71,122]. This can be attributed to the possibility of the SCM reacting with the portlandite of the RA in the interface zone, which is the most porous zone of the system.

Results suggest that the use of RA does not have a unique effect on ESA. RAs have some positive effects that can compensate for the negative effect of higher transport properties, and this effect depends on other test parameters. On the other hand, the results indicate robust control of the quality of the matrix in the development of the ESA. Concretes using SCM as a cement substitute had improved performance.

#### 3.1.2. Recycled Aggregate Mortar (RAM)

Figure 2 shows the degradation ratio for the case of FRA mortars exposed to pure ESA. In this case, there are very few results in the literature, and the observed spread is wide.



Figure 2. Degradation ratios of mortars with different percentages of FRA exposed to pure ESA [59,60,69,87,91,94].

Lee [59] and Lee et al. [60] report a positive effect of replacing up to 50% of the fine natural aggregate with FRA in sodium or magnesium sulfate solutions, showing lower expansion. Moreover, RAMs with 75 and 100% show higher damage than control mortar. On the contrary, Santillán et al. [69] report higher damage for mortar with 30% FRA than for mortar with 50% FRA, in terms of expansion and weight variation. The different experimental setups of each study can explain this difference between them. For example, no water correction for FRA absorption is reported in Lee [59] or Lee et al. [60], and so the effective w/c ratio could be lower for low replacement values. On the other hand, in Santillan et al. [69], additional water was added to the dosage to compensate for FRA absorption. However, the determination of this value is not accurate and involves significant error [137].

Although there are few results in the literature, they show a predominant effect of the quality of the RAM matrix on their sulfate resistance, as in the case of RAC. Boudali et al. [91] and Candamano et al. [94] evaluated RAM with higher strength levels (self-compacting) and 100% FRA. The results show similar or even better performance for RAM than for the reference, especially when using an SCM in the composition.

As with RAC, RAM performance shows a possible positive effect of FRA against ESA, which, depending on other parameters, can compensate for the negative effect of the higher porosity of this aggregate.

#### 3.2. ESA + PSA

Few studies have evaluated the performance of RAC under exposure conditions that promote PSA. Figure 3 shows the results collected from the literature.



**Figure 3.** Degradation ratios of concretes with different % CRA exposed to ESA + PSA [63,65,77,83,84,88,89,91].

Most of the cited works compare conventional concretes to concretes with 100% RA [55,63,77,84,89,91]. Most results show worse performance for RAC than for NAC under this type of exposure, with the degradation ratio ranging from 1.04 to 1.39. The only case where better results were obtained was in Boudali et al. [91], where self-compacting concretes with 100% RA and different cementitious compositions (replacement by SCM) were studied. In this case, significantly lower degradation ratios were observed in the RAC than in the standard, with the SCM having a significant effect.

Only three papers evaluate the performance of concretes with partial RA replacement, namely, the studies by Zega et al. [83], Qi et al. [65], and Al-Baghdadi [68]. As described for ESA, the results do not show a consistent trend in the relationship between RA content and deterioration, which can be attributed to the different setups of each campaign. In addition, better performance of RAC is sometimes observed with respect to the reference concrete, again showing the possible positive effect of RA in some cases.

In Zega et al. [83], two series of concretes were evaluated: Series 1 with w/c ratio = 0.50 and Series 2 with w/c ratio = 0.38. For the first series, after ten years of exposure to saline soil (semi-buried), the decrease in the modulus of elasticity of the concretes with 25 and 75% was higher than that of the reference concrete. On the other hand, for the second series, practically no differences were observed between the NAC and the RAC, showing that the matrix quality can control the possible adverse effect of the RA.

RAC tested by Qi et al. [65] was subjected to wetting and drying cycles in a sodium sulfate solution. In this case, smaller decreases in the modulus of elasticity of RAC than those of the reference concretes were observed for those made with 30 and 50% RA and larger decreases for concretes with 70 and 100% RA. These results are confirmed by the sulfate ingress profiles calculated by analytical chemistry. The authors attribute this result to a slight change in the effective w/c ratio due to the moisture state of the aggregates.

Finally, Al-Baghdadi [68] used concretes made with RA pretreated by saturation in a polyvinyl alcohol (PVA) solution and subjected to saturation and drying cycles with a magnesium sulfate solution. In this case, the concretes obtained better results as the inclusion of RA increased, and the authors attribute these results to an internal curing effect generated by the RA and promoted by the addition of PVA. As in the case of ESA, the results regarding ESA + PSA are not enough to establish consistent correlations, but they do confirm some points regarding the development of the sulfate attack in recycled concretes, such as the quality of the new matrix, the ambiguous effect of RA, and the very positive effect of SCM with RA.

#### 3.3. ISA

As mentioned above, the ISA issue becomes relevant for RA due to the potential for contamination with sulfate-rich building materials, mainly gypsum. Most regulations prescribe the limiting content of sulfates in aggregates as a total percentage of their weight, with values between 0.2 and 1.0% [9,11]. However, several studies have investigated concretes and mortars with recycled aggregates contaminated with sulfate-based materials, and the results show that these limits may be conservative.

A summary of the reviewed papers is presented in Table 1. Several studies evaluate the effect of the sulfate content in RA [46,48,73,80,81]. For example, Tovar-Rodriguez et al. [73] show that mortars with 4.3% SO<sub>3</sub> content present three to four times greater expansion than mortars with 2.9% SO<sub>3</sub> content, even when an SRPC is used. The authors also demonstrate that 100% FRA mortar performs adequately in terms of durability, even with a 2.9% SO<sub>3</sub> content, which exceeds regulatory limitations. They base this conclusion on a mathematical prediction of service-life expansion. Agrela et al. [46] investigate samples of cement-treated granular material (low cement content and mechanical compaction) with 100% RA, different cement types, and different levels of gypsum contamination. They show that by using an SRPC, the SO<sub>3</sub> content of the aggregate can be increased up to 1.3% without any risk of ISA failure.

Paper	Mixes	Sulfate	Highlights
Tovar-Rodriguez et al. [73]	RAM w/c = 0.35 100% FRA OPC and SRPC	FRA with $SO_3 = 2.9\%$ a.s. Contamination with different sulfate sources to increase $SO_3$ content to 4.3% bwa.	<ul> <li>Mortars with 100% of FRA with 2.9% SO<sub>3</sub> perform well in terms of durability (higher than regulation limits).</li> <li>SO<sub>3</sub> bwa content of the RAMs has a much higher influence than the cement sulfate resistance.</li> </ul>
Agrela et al. [46]	CTGM 100% CRA OPC and SRPC	CRA with SO <sub>3</sub> = 0.3% a.s. Contamination with hydrated gypsum to increase SO <sub>3</sub> content to 1.0 and 2.5% bwa.	• The limiting SO <sub>3</sub> content of the aggregate can be increased if SRPC is used in the mix (up to 1.3%).
Colman [47]	RAM w/c = 0.35 to 0.65 100% FRA OPC and SRPC	FRA with $SO_4 = 0.18\%$ w.s. Contamination with different size gypsum grains to increase $SO_4$ content to 0.47% and 3.08% bwa.	<ul> <li>The expansion of RAMs increases with SO<sub>3</sub> content.</li> <li>Other set parameters, such as RA temperature or alkalinity, have a noticeable effect on RAMs' behavior.</li> </ul>
Abid et al. [45]	RAC w/c = 0.55 0 to 100% CRA	CRA contaminated with laboratory-made plaster. SO <sub>3</sub> content not reported.	• Greater expansion occurs in concretes with higher CRA content.
Colman et al. [48]	RAM 0 and 100% FRA	RAMs with NA and FRAs with $SO_3 = 0.08\%$ to 0.62%. Contamination with gypsum increases the $SO_3$ content to 0.47% and 3.08% bwa.	<ul> <li>The FRA mortars perform similarly to the control mortar despite their SO<sub>3</sub> content of up to 0.68%.</li> <li>The mortar contaminated with more than 3% SO<sub>3</sub> show a significantly higher degree of expansion.</li> </ul>

Table 1. Summary of papers on ISA in RAC.

Paper	Mixes	Sulfate	Highlights
Yammine et al. [81]	RAM (heat curing) w/c = 0.50 0 and 100% FRA OPC	Mortars with NA and two FRAs with $SO_3 = 0.54\%$ and $0.65\%$ a.s., respectively. Contamination with sodium sulfate increases the $SO_3$ content to 5% bwc, and contamination with plaster powder increases the $SO_3$ content to 0.80% and 1.20% bwa.	<ul> <li>Among contaminated mortars, better performance is reported for mortars with FRA than for the reference mortar.</li> <li>Mortars in which FRA is contaminated with 1.2% SO<sub>3</sub> show similar expansion to the control.</li> </ul>
Yammine et al. [80]	RAC (heat curing) w/c = 0.50 0 and 100% CRA OPC and SCM	Concretes with NA and two CRAs with $SO_4 = 0.11\%$ and $0.29\%$ w.s., respectively. Contamination with sodium sulfate increases the $SO_4$ content to 2.7% bwc.	• Better performance is observed in concretes with recycled aggregate than in reference concrete.

Table 1. Cont.

CTGM: cement-treated granular material; a.s.: acid-soluble; w.s.: water-soluble; bwa: by weight of aggregate; bwc: by weight of cement.

Colman et al. [48] evaluate mortars with contaminated FRA and different set-up parameters, such as temperature, alkalinity, and gypsum source. Mortars with higher SO<sub>3</sub> content showed higher expansion, but other parameters showed a considerable effect, such as temperature, sulfate resistance of cement, and alkalinity (due to CH lixiviation of RAs). Colman et al. [47] evaluate RAMs using several commercial FRAs with sulfate contents ranging from 0.08% to 0.62% and mixtures contaminated with gypsum (up to 3.08% SO<sub>3</sub>). Mortars with commercial FRAs showed similar behavior to control mortar with NA, while mortar with 3.08% SO<sub>3</sub> showed considerably greater expansion. The authors suggest that some regulations may have conservative prescriptions (0.2% in the authors' location).

Other studies have shown the effects of RA on the development of ISA. Yammine et al. [81] evaluate mortars with and without RA from two different sources contaminated with sodium sulfate. These mortars were subjected to temperature curing, which is common in the precast industry and may promote the subsequent development of DEF (delayed ettringite formation). In this case, slightly greater expansion was found in the uncontaminated samples for the FRA mortars. At the same time, a strong positive effect of using FRA was observed in the contaminated samples. Microstructural analyses showed that the mortars with FRA had greater incorporation of air and, consequently, a higher number of small air bubbles (20 to 200 µm) acting as crystallization points without associated confinement. Once again, these results highlight the ambiguous effect of porosity on deterioration processes involving the formation of expansion products [100,142,143]. For example, Colman et al. [47] show greater expansion for mortar with a lower w/c ratio and, hence, lower porosity. Yammine et al. [80] evaluate the expansion of NAC and RAC cured at elevated temperatures. They report lower expansion for RAC than for NAC, which is attributed to the lower internal constraint that can be achieved using recycled aggregates. Moreover, the results of some papers [47,80] show that higher alkalinity for RA can improve the performance of mixes against the ISA, and it should be considered in further experimental setups. In the case of Abid et al. [45], concretes with contaminated RAC are evaluated for ISA. The results show an increased loss of mechanical strength with increasing RA content. However, only one level of sulfate contamination was used (not reported well), and so it cannot be confirmed whether the increased deterioration is an effect of the RA or the associated increased gypsum contamination.

Studies show that the aggregate sulfate limits set by regulations may be conservative. However, more experimental results are needed to better understand the effects of RA on the ISA under different conditions.

#### 4. Discussion

#### 4.1. Effect of RA

Our literature review shows that RAs can affect the performance of cementitious mixtures exposed to sulfate attack. The general trend shows worse performance in RACs than in NACs, mainly due to a higher porosity that accelerates the kinetics of the degradation mechanisms. Several hypotheses attempt to explain this phenomenon. Some authors point out that the higher porosity of RAC results in a lower internal restraint to deformation (lower modulus of elasticity), which delays crack propagation initiation [62,65,67,94]. The better strain compatibility of RA than NA with the cement matrix has previously been reported by Corinaldesi et al. [131]. Yammine et al. [81] attribute the lower constraint to the higher amount of entrapped air in RAC mixes [126], which creates mesopores that can act as preferential unconfined crystallization spaces for the ESA products. Another less consistent hypothesis is that the expansive products can precipitate in the pores of the RA, resulting in lower internal stresses [60]. However, there is no conclusive evidence that ESA products precipitate preferentially in the pores of the RAC. Regardless of which hypothesis is correct, the main difficulty with this effect is that it follows different patterns for different experimental campaigns. This suggests that other test parameters also influence the relationship between the negative and positive effects of porosity (e.g., materials used, exposure conditions, evaluation method).

The results show that in most cases, the performance of RAC depends more on the physical–chemical quality of the new matrix than on the quantity of RA included. In many works, an almost null effect of RA is observed in mixes with a low w/c ratio or with SR-cement, even for large replacements with RA. Instead, the effects of RA are observed most in concretes of lower quality. An important aspect related to porosity is that RA can modify the effective w/c ratio in its surroundings if the moisture compensation during the mix design and elaboration stages is incorrect [23,40,136,137,144]. This can lead to experimental errors, even with partial replacement by RA, as we would be adding or removing free water to or from the concrete during its hydration process. This difficulty is more significant in the case of FRA due to the difficulty of measuring water absorption in this fraction [145].

For ESA, studies have not yet evaluated the effects of the physical–chemical quality of the RA. The number of studies is very small, and the characterization of the RA is insufficient to draw sound conclusions. The two most important properties of RA that may affect ESA mechanisms are the porosity and the reactive phase content in its cement paste (AFm, CH, and C-S-H). The porosity of RA depends, as mentioned above, on the mortar/paste content, the quality of the original concrete, and the crushing process, among other factors. The mineral composition of the RA, specifically the cement paste, depends on the type of cement in the source concrete and the degree of carbonation of the RA. For ISA, the mineral composition is also strongly influenced by contamination, particularly sulfate-based materials.

The characterization of RA must be accompanied by a study of their effect on RAC. It is interesting to know whether RA's potential reactivity becomes effective under certain exposure conditions or under different design criteria. Further investigations are needed to determine whether the contribution of reactive phases may have an additional effect. In addition, knowledge about the ambiguous effect of RA observed in several studies should be deepened. The results achieved in relation to the aforementioned premises can lead to accurate classification protocols for RA and design specifications for recycled mixes.

#### 4.2. Sulfate-Resistant RAC Design

Although the results analyzed are insufficient to fully explain the effect of RA on sulfate performance, they provide information about technological solutions that may improve the performance of RAC against sulfates. The literature review highlights the significance of the RAC matrix on the potential effects of RA. The results suggest that recycled concretes, designed using conventional concrete criteria (low w/c, SRPC), exhibit satisfactory performance even with substantial aggregate replacements. The discussion

regarding admissible percentages in regulations and the potential impact of RA properties on these limits requires further research to draw reliable conclusions.

Another important discussion is about the limitation of sulfate content by RA to avoid the development of ISA. Existing studies indicate that the current limits established for natural aggregates are quite conservative. The existing limits for  $SO_3$  content in aggregates, which range between 0.2% and 1.0%, could be further increased, especially if additional design criteria, such as the use of SRPC, are adopted. However, due to the scarcity of literature, reliable values for regulation cannot be defined.

In addition, many studies provide strategies to optimize the performance of RAC, among which three groups can be highlighted: the use of SCMs, RAC procedural improvements, and pretreatment of RA. The use of pozzolanic SCMs as a strategy to increase the resistance of concrete to sulfate attack has shown equal or even superior efficiency in RACs compared to their respective conventional concretes (NACs) [45,54,62,65,66,71,72,78,90,91,146,147]. This is extremely important because it means that eco-efficient concretes (such as those with RA and a lower clinker factor) can perform as well as or better than comparable conventional concretes in terms of resistance to sulfate attack. This synergistic combination of RA and SCM has been highlighted by several authors and linked to other concrete properties [89,127,130,138,148,149].

Some authors propose improvements in RAC through strategies in the design and processing procedure. For example, the two-stage mixing approach improves the mechanical properties of RAC, mainly explained by a substantial improvement in the ITZ [130,138,150], a fundamental phase in the transport properties of hardened concrete [36,138]. Other works point to design strategies. Bui et al. [151] suggest using only a fraction of the RA (the coarser fraction), allowing the replacement percentages to be increased without losing performance.

Finally, many studies evaluate different pretreatment methods for RA to improve overall durability performance [15,152,153], including resistance to sulfate attack [56,68,70,154]. Existing pretreatment techniques for RA are divided into attached-mortar improvement techniques and attached-mortar removal techniques [153]. Among the first, different cementitious slurries were tested by Shayan et al. [70] and Zhang et al. [154], showing a positive effect on RAC sulfate resistance. In another case, Kazmi et al. [56] evaluated the sulfate resistance of mixes containing RA pretreated by accelerated carbonation, highlighting the improvement in the quality of RA and the recapture of  $CO_2$  from the atmosphere [22,56]. Al-Baghdidi et al. [68] evaluated the effect of pretreatment with a polyvinyl alcohol (PVA) solution. The results showed an increase in RAC sulfate resistance with increasing RA content. In this case, the PVA impregnation acted as a water reservoir, providing the conditions for internal curing. This potential characteristic of RA as an internal curing agent has also been studied with positive results and is a path to be developed [136,138].

#### 4.3. Guidelines for Further Research

Based on the information collected and the analysis carried out, general guidelines are proposed for further research to fully understand the effect of RA on the sulfate-attack resistance of RAC. Table 2 summarizes the key points of this section.

Firstly, RA must be rigorously characterized at the physical and chemical level to assess their hygroscopic and thermodynamic behavior correctly [135,137,155]. The most suitable parameters for standardization are mortar content (cement paste for FRA), porosity, absorption, and paste composition (chemical and mineralogical) [44,134,156,157].

Regarding mixture design, different levels and qualities of RAC should be considered. For RAC performance testing, standard criteria should be established for matrix quality, evaluation methods, and exposure conditions. Matrix quality should be approached by varying the w/c ratio, changing the type of cement, using SCMs (different types and contents), and considering different mixture designs. In terms of measurement, in addition to the traditional methods used (nondestructive and destructive tests), the proposal of new methods or methods adapted from other tests may be necessary. The most useful approach is that RACs should comply with regulations' prescriptions for conventional concretes exposed to the same conditions, which are usually focused on a low w/c ratio and the use of SR binder.

Table 2. Test parameters to consider in further experimental campaigns.

Level	Group	Variables	
Aggregate	Physical properties	<ul><li>Porosity</li><li>Absorption</li><li>Attached mortar/paste content</li></ul>	
characterization	Chemical properties	<ul> <li>Phase quantification (CH, AFm)</li> <li>Sulfate content (SO<sub>3</sub>)</li> <li>Alkalinity</li> </ul>	
	Matrix-related parameters	<ul> <li>Matrix porosity (w/c ratio)</li> <li>Cement type (SRPC and others), SCMs, and admixtures</li> </ul>	
Performance tests on concrete and mortar	Aggregate related parameters	<ul><li>RA replacement</li><li>RA's physical and chemical properties</li></ul>	
samples	Exposure related parameters	<ul> <li>ISA: sulfate contamination, alkalinity</li> <li>Saturation condition (ESA or ESA + PSA), sulfate media concentration</li> </ul>	

In terms of exposure, accelerated tests are usually used, the concentrations of which generally fall within the most aggressive conditions of the regulations. In the case of saturated condition tests, this criterion is more justified, as diffusion transport makes ESA a process of prolonged deterioration over time. In the case of wetting–drying cycle tests, some works in the literature evaluate protocols and concentrations that provide better test times and have some correlation with long-term tests [117,158]. For partial saturation tests, there is no single criterion, but in general, these tests use semi-buried or semi-submerged specimens in soils or sulfate solutions, with concentrations that are also very aggressive [57,88,159]. Regardless of the method applied, the most relevant aspect to delve into is finding information that includes the results of accelerated tests and field exposures.

In the case of ISA, the published results show that the regulations should consider the case of RA, especially regarding the limited content of soluble SO<sub>3</sub>. Further studies can be carried out to better adjust the allowable limit according to other parameters of the mix, such as the type of cement, alkalinity, and strength level of the new material. It is also interesting to know the effect of temperature curing [85,86], which is a common practice in the precast industry. In addition, the effect of different types of contaminants on the RA should be evaluated [55].

Finally, the potential of simulating degradation processes by formulating numerical mathematical models [96,160–162] is worth mentioning. The main advantage of these methods is that they significantly reduce the logistical and operational costs of testing, especially when many variables need to be evaluated, as in the case of sulfate attack. However, the numerical models themselves require traditional tests to be calibrated. This area of research is still under development, but its potential is very promising.

#### 5. Conclusions

This review summarizes the performance of cementitious mixture with recycled concrete aggregates (RA) against different types of sulfate attack. Existing studies about this topic were compiled, and their results were analyzed in terms of degradation ratio and RAC content. Based on the above, the following conclusions can be drawn:

• The performance of RAC against sulfate attack is strongly dependent on the properties of the new cement matrix, with well-designed recycled concretes demonstrating good performance even with substantial amounts of RA as replacements.

- Using RA generally decreases concrete's resistance to sulfate attack due to increased transport properties. However, in some cases, RA can mitigate the negative effects by compensating for increased porosity.
- It remains unclear whether RA can significantly increase the quantity of sulfate-reactive phases, necessitating further studies with detailed characterization of RA and concrete performance tests using the same RA.
- The combination of RA with SCM can have a certain synergy of positive effects, resulting in even higher eco-efficiency for the material. Various pretreatment techniques for RA can also increase the sulfate resistance of RAC.
- The limits for SO<sub>3</sub> regulations should be adjusted for recycled aggregate, as the current limits for natural aggregates may be inadequate for RA. These limits could be even higher if special precautions are taken in concrete design, such as using SR cement or SCM.

Further research should explore aspects such as internal restraint and potential reactivity of RA. The development of a standard method, including test conditions and evaluation parameters, would greatly benefit the study of different types, qualities, and quantities of RA in cementitious mixtures.

**Author Contributions:** Conceptualization, L.R.S., C.J.Z. and E.F.I.; formal analysis, L.R.S., C.J.Z. and E.F.I.; investigation, L.R.S.; data curation, L.R.S.; writing—original draft preparation, L.R.S.; writing—review and editing, L.R.S., C.J.Z. and E.F.I.; funding acquisition, C.J.Z. and L.R.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research was partially funded by FONCyT, grant number PICT 2015-3339 Préstamo BID.

**Data Availability Statement:** The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflicts of interest.

#### Nomenclature

CDW	construction and demolition waste	ESA	external sulfate attack
RA	recycled concrete aggregates	ISA	internal sulfate attack
RAC	recycled aggregate concrete	PSA	physical sulfate attack
RAM	recycled aggregate mortar	CH	calcium hydroxide
NA	natural aggregates	SCM	supplementary cementitious materials
NAC	natural aggregate concrete	ITZ	interfacial transition zone
CRA	Coarse Recycled Aggregate	OPC	Ordinary Portland Cement
FRA	Fine Recycled Aggregate	SRPC	Sulfate-resistant Portland

#### References

- 1. Cordoba, G.; Paulo, C.I.; Irassar, E.F. Towards an Eco-Efficient Ready Mix-Concrete Industry: Advances and Opportunities. A Study of the Metropolitan Region of Buenos Aires. *J. Build. Eng.* **2023**, *63*, 105449. [CrossRef]
- 2. Kadawo, A.; Sadagopan, M.; During, O.; Bolton, K.; Nagy, A. Combination of LCA and Circularity Index for Assessment of Environmental Impact of Recycled Aggregate Concrete. *J. Sustain. Cem. Mater.* **2023**, *12*, 1–12. [CrossRef]
- 3. Mehta, K.P. Reducing the Environmental Impact of Concrete. *Concr. Int.* **2001**, 23, 61–66.
- 4. Coelho, A.; De Brito, J. Influence of Construction and Demolition Waste Management on the Environmental Impact of Buildings. *Waste Manag.* 2012, 32, 532–541. [CrossRef]
- 5. Abbas, A.; Fathifazl, G.; Isgor, O.B.; Razaqpur, A.G.; Fournier, B.; Foo, S. Environmental Benefits of Green Concrete. In Proceedings of the 2006 IEEE EIC Climate Change Conference, Otawa, ON, Canada, 10–12 May 2006; Volume 145, pp. 1–3. [CrossRef]
- 6. Akhtar, A.; Sarmah, A.K. Construction and Demolition Waste Generation and Properties of Recycled Aggregate Concrete: A Global Perspective. *J. Clean. Prod.* **2018**, *186*, 262–281. [CrossRef]
- Rao, A.; Jha, K.N.; Misra, S. Use of Aggregates from Recycled Construction and Demolition Waste in Concrete. *Resour. Conserv. Recycl.* 2007, 50, 71–81. [CrossRef]
- 8. Senaratne, S.; Lambrousis, G.; Mirza, O.; Tam, V.W.Y.; Kang, W.H. Recycled Concrete in Structural Applications for Sustainable Construction Practices in Australia. *Procedia Eng.* **2017**, *180*, 751–758. [CrossRef]

- Tam, V.W.Y.; Soomro, M.; Evangelista, A.C.J. A Review of Recycled Aggregate in Concrete Applications (2000–2017). Constr. Build. Mater. 2018, 172, 272–292. [CrossRef]
- 10. Shrivastava, S.; Chini, A. Construction Materials and C&D Waste in India. Lifecycle Des. Build. Syst. Mater. 2009, 1–152.
- Gonçalves, P.; De Brito, J. Recycled Aggregate Concrete (RAC)—Comparative Analysis of Existing Specifications. *Mag. Concr. Res.* 2010, *62*, 339–346. [CrossRef]
- 12. Jin, R.; Chen, Q. Overview of Concrete Recycling Legislation and Practice in the United States. J. Constr. Eng. Manag. 2019, 145, 05019004. [CrossRef]
- 13. Duan, H.; Wang, J.; Huang, Q. Encouraging the Environmentally Sound Management of C&D Waste in China: An Integrative Review and Research Agenda. *Renew. Sustain. Energy Rev.* **2015**, *43*, 611–620. [CrossRef]
- 14. Kwan, W.H.; Ramli, M.; Kam, K.J.; Sulieman, M.Z. Influence of the Amount of Recycled Coarse Aggregate in Concrete Design and Durability Properties. *Constr. Build. Mater.* **2012**, *26*, 565–573. [CrossRef]
- 15. Dimitriou, G.; Savva, P.; Petrou, M.F. Enhancing Mechanical and Durability Properties of Recycled Aggregate Concrete. *Constr. Build. Mater.* **2018**, *158*, 228–235. [CrossRef]
- 16. Di Maio, A.A.; Giaccio, G.; Zerbino, R. Hormigones Con Agregados Reciclados. Cienc. Tecnol. Hormigón 2002, 9, 5–10.
- 17. Bravo, M.; De Brito, J.; Pontes, J.; Evangelista, L. Mechanical Performance of Concrete Made with Aggregates from Construction and Demolition Waste Recycling Plants. J. Clean. Prod. 2015, 99, 59–74. [CrossRef]
- 18. Gómez-Soberón, J.M. Porosity of Recycled Concrete with Substitution of Recycled Concrete Aggregate. An Experimental Study. *Cem. Concr. Compos.* **2002**, *32*, 1301–1311. [CrossRef]
- Kou, S.C.; Poon, C.S.; Chan, D. Influence of Fly Ash as Cement Replacement on the Properties of Recycled Aggregate Concrete. J. Mater. Civ. Eng. 2007, 19, 709–717. [CrossRef]
- 20. De Brito, J.; Ferreira, J.; Pacheco, J.; Soares, D.; Guerreiro, M. Structural, Material, Mechanical and Durability Properties and Behaviour of Recycled Aggregates Concrete. *J. Build. Eng.* **2016**, *6*, 1–16. [CrossRef]
- 21. Xiao, J.; Li, L.; Tam, V.W.Y.; Li, H. The State of the Art Regarding the Long-Term Properties of Recycled Aggregate Concrete. *Struct. Concr.* **2014**, *15*, 3–12. [CrossRef]
- 22. Guo, H.; Shi, C.; Guan, X.; Zhu, J.; Ding, Y.; Ling, T.C.; Zhang, H.; Wang, Y. Durability of Recycled Aggregate Concrete—A Review. *Cem. Concr. Compos.* **2018**, *89*, 251–259. [CrossRef]
- 23. González-Fonteboa, B.; Seara-Paz, S.; De Brito, J.; González-Taboada, I.; Martínez-Abella, F.; Vasco-Silva, R. Recycled Concrete with Coarse Recycled Aggregate. An Overview and Analysis. *Mater. Constr.* **2018**, *68*, e151. [CrossRef]
- Ma, Z.; Tang, Q.; Yang, D.; Ba, G. Durability Studies on the Recycled Aggregate Concrete in China over the Past Decade: A Review. Adv. Civ. Eng. 2019, 2019, 4073130. [CrossRef]
- 25. Zega, C.J.; Santillán, L.R.; Sosa, M.E.; Villagrán Zaccardi, Y.A. Durable Performance of Recycled Aggregate Concrete in Aggressive Environments. *J. Mater. Civ. Eng.* **2020**, *32*, 03120002. [CrossRef]
- 26. Skalny, J.; Marchand, J.; Odler, I. Sulfate Attack on Concrete; Spon: London, UK, 2002; ISBN 0-203-34224-0.
- 27. Trägårdh, J.; Bellmann, F. Sulphate Attack. RILEM Rep. 2007, 38, 89–118. [CrossRef]
- Mehta, K.P. Mechanism of Sulfate Attack on Portland Cement Concrete—Another Look. Cem. Concr. Res. 1983, 13, 401–406. [CrossRef]
- Irassar, E.F.; Di Maio, Á.A.; Batic, O.R. Sulfate Attack on Concrete with Mineral Admixtures. Cem. Concr. Res. 1996, 26, 113–123. [CrossRef]
- 30. Haynes, H.; O'Neill, R.; Mehta, K.P. Concrete Deterioation from Physical Attack by Salts. Concr. Int. 1996, 18, 63–68.
- 31. Santhanam, M.; Cohen, M.D.; Olek, J. Sulfate Attack Research—Whither Now? Cem. Concr. Res. 2001, 31, 845–851. [CrossRef]
- Menéndez-Mendez, E.; Matschei, T.; Glasser, F.P. Sulfate Attack of Concrete. In *Performance of Cement-Based Materials in Aggressive Aqueous Environments*; Alexander, M., Bertron, A., De Belie, N., Eds.; Springer: Dordrecht, The Netherlands, 2013; pp. 8–74, ISBN 9789400754133.
- 33. Whittaker, M.; Black, L. Current Knowledge of External Sulfate Attack. Adv. Cem. Res. 2015, 27, 532–545. [CrossRef]
- Liu, Z.; Hu, W.; Hou, L.; Deng, D. Effect of Carbonation on Physical Sulfate Attack on Concrete by Na<sub>2</sub>SO<sub>4</sub>. Constr. Build. Mater. 2018, 193, 211–220. [CrossRef]
- Elahi, M.M.A.; Shearer, C.R.; Naser Rashid Reza, A.; Saha, A.K.; Khan, M.N.N.; Hossain, M.M.; Sarker, P.K. Improving the Sulfate Attack Resistance of Concrete by Using Supplementary Cementitious Materials (SCMs): A Review. *Constr. Build. Mater.* 2021, 281, 122628. [CrossRef]
- Xiao, J.; Li, W.; Sun, Z.; Lange, D.A.; Shah, S.P. Properties of Interfacial Transition Zones in Recycled Aggregate Concrete Tested by Nanoindentation. *Cem. Concr. Compos.* 2013, 37, 276–292. [CrossRef]
- Olorunsogo, F.T.; Padayachee, N. Performance of Recycled Aggregate Concrete Monitored by Durability Indexes. *Cem. Concr. Res.* 2002, 32, 179–185. [CrossRef]
- Gonçalves, A.; Esteves, A.; Vieira, M. Influence of Recycled Concrete Aggregates on Concrete Durability. In Proceedings of the International RILEM Conference "The Use of Recycled Materials in Building and Structures", Barcelona, Spain, 8–11 November 2004; Vázquez, E., Hendriks, C.F., Janssen, G.M.T., Eds.; pp. 554–562.
- 39. Abbas, A.; Fathifazl, G.; Isgor, O.B.; Razaqpur, A.G.; Fournier, B.; Foo, S. Durability of Recycled Aggregate Concrete Designed with Equivalent Mortar Volume Method. *Cem. Concr. Compos.* **2009**, *31*, 555–563. [CrossRef]

- 40. Evangelista, L.; de Brito, J. Durability Performance of Concrete Made with Fine Recycled Concrete Aggregates. *Cem. Concr. Compos.* **2010**, *32*, 9–14. [CrossRef]
- 41. Zega, C.J.; Di Maio, Á.A. Use of Recycled Fine Aggregate in Concretes with Durable Requirements. *Waste Manag.* 2011, 31, 2336–2340. [CrossRef]
- Thomas, C.; Setién, J.; Polanco, J.A.; Alaejos, P.; Sánchez De Juan, M. Durability of Recycled Aggregate Concrete. *Constr. Build. Mater.* 2013, 40, 1054–1065. [CrossRef]
- 43. Saravanakumar, P.; Dhinakaran, G. Durability Aspects of HVFA-Based Recycled Aggregate Concrete. *Mag. Concr. Res.* 2014, 66, 186–195. [CrossRef]
- 44. Bravo, M.; De Brito, J.; Pontes, J.; Evangelista, L. Durability Performance of Concrete with Recycled Aggregates from Construction and Demolition Waste Plants. *Constr. Build. Mater.* **2015**, *77*, 357–369. [CrossRef]
- 45. Abid, S.R.; Nahhab, A.H.; Al-aayedi, H.K.H.; Nuhair, A.M. Expansion and Strength Properties of Concrete Containing Contaminated Recycled Concrete Aggregate. *Case Stud. Constr. Mater.* **2018**, *9*, e00201. [CrossRef]
- Agrela, F.; Cabrera, M.; Galvín, A.P.; Barbudo, A.; Ramirez, A. Influence of the Sulphate Content of Recycled Aggregates on the Properties of Cement-Treated Granular Materials Using Sulphate-Resistant Portland Cement. *Constr. Build. Mater.* 2014, 68, 127–134. [CrossRef]
- 47. Colman, C.; Bulteel, D.; Thiery, V.; Rémond, S.; Michel, F.; Courard, L. Internal Sulfate Attack in Mortars Containing Contaminated Fine Recycled Concrete Aggregates. *Constr. Build. Mater.* **2021**, 272, 121851. [CrossRef]
- 48. Colman, C.; Bulteel, D.; Rémond, S.; Zhao, Z.; Courard, L. Valorization of Fine Recycled Aggregates Contaminated with Gypsum Residues: Characterization and Evaluation of the Risk for Secondary Ettringite Formation. *Materials* **2020**, *13*, 4866. [CrossRef]
- Corral-Higuera, R.; Arredondo-Rea, S.P.; Neri-Flores, M.A.; Gómez-Soberón, J.M.; Almeraya Calderón, F.; Castorena-González, J.H.; Almaral-Sánchez, J.L. Sulfate Attack and Reinforcement Corrosion in Concrete with Recycled Concrete Aggregates and Supplementary Cementing Materials. *Int. J. Electrochem. Sci.* 2011, 6, 613–621. [CrossRef]
- 50. De Schepper, M.; Van den Heede, P.; Arvaniti, E.C.; De Buysser, K.; Van Driessche, I.; De Belie, N. Sulfates in Completely Recyclable Concrete and the Effect of CaSO4 on the Clinker Mineralogy. *Constr. Build. Mater.* **2017**, *137*, 300–306. [CrossRef]
- 51. Debieb, F.; Courard, L.; Kenai, S.; Degeimbre, R. Mechanical and Durability Properties of Concrete Using Contaminated Recycled Aggregates. *Cem. Concr. Compos.* 2010, *32*, 421–426. [CrossRef]
- 52. Dhir, R.K.; Limbachiya, M.C.; Leelawat, T. Suitability of Recycled Concrete Aggregate for Use in Bs 5328 Designated Mixes. *Proc. Inst. Civ. Eng.-Struct. Build.* **1999**, 134, 257–274. [CrossRef]
- 53. Huang, Y.; Wang, P.; Yang, H.; Wang, Y.; Li, L. Wind Erosion of Recycled Concrete on Sulfate Attack. In *IOP Conference Series: Materials Science and Engineering*; IOP Publishing: Bristol, UK, 2018; Volume 452. [CrossRef]
- Hwang, J.P.; Shim, H.B.; Lim, S.; Ann, K.Y. Enhancing the Durability Properties of Concrete Containing Recycled Aggregate by the Use of Pozzolanic Materials. *KSCE J. Civ. Eng.* 2013, 17, 155–163. [CrossRef]
- 55. Jia, P.; Li, L.; Zhou, J.; Zhang, D.; Guan, Z.; Dong, J.; Wang, Q. Performance Evolution of Recycled Aggregate Concrete under the Coupled Effect of Freeze–Thaw Cycles and Sulfate Attack. *Appl. Sci.* **2022**, *12*, 6950. [CrossRef]
- 56. Kazmi, S.M.S.; Munir, M.J.; Wu, Y.F.; Patnaikuni, I.; Zhou, Y.; Xing, F. Effect of Different Aggregate Treatment Techniques on the Freeze-Thaw and Sulfate Resistance of Recycled Aggregate Concrete. *Cold Reg. Sci. Technol.* **2020**, *178*, 103126. [CrossRef]
- Al-Attar, T.S.; Al-Khateeb, A.M.; Bachai, A.H. Behavior of High Performance Concrete Exposed to Internal Sulfate Attack (Gypsum-Contaminated Aggregate). In Proceedings of the Earth & Space 2006: Engineering, Construction, and Operations in Challenging Environment, League City/Houston, TX, USA, 5–8 March 2006; Volume 161. [CrossRef]
- 58. Kotwal, S.; Singh, H.; Kumar, R. Materials Today: Proceedings Study on Sulphate and Chloride Resistance of Self-Compacting Concrete. *Mater. Today Proc.* 2021, 48, 1044–1047. [CrossRef]
- 59. Lee, S. Influence of Recycled Fine Aggregates on the Resistance of Mortars to Magnesium Sulfate Attack. *Waste Manag.* 2009, 29, 2385–2391. [CrossRef]
- Lee, S.; Swamy, R.N.; Kim, S.; Park, Y. Durability of Mortars Made with Recycled Fine Aggregates Exposed to Sulfate Solutions. J. Mater. Civ. Eng. 2008, 20, 63–70. [CrossRef]
- 61. Lei, B.; Li, W.; Tang, Z.; Tam, V.W.Y.; Sun, Z. Durability of Recycled Aggregate Concrete under Coupling Mechanical Loading and Freeze-Thaw Cycle in Salt-Solution. *Constr. Build. Mater.* **2018**, *163*, 840–849. [CrossRef]
- 62. Li, Y.; Wang, R.; Li, S.; Zhao, Y.; Qin, Y. Resistance of Recycled Aggregate Concrete Containing Low- and High-Volume Fly Ash against the Combined Action of Freeze–Thaw Cycles and Sulfate Attack. *Constr. Build. Mater.* **2018**, *166*, 23–34. [CrossRef]
- 63. Li, Y.; Yang, X.; Lou, P.; Wang, R.; Li, Y.; Si, Z. Sulfate Attack Resistance of Recycled Aggregate Concrete with NaOH-Solution-Treated Crumb Rubber. *Constr. Build. Mater.* **2021**, *287*, 123044. [CrossRef]
- 64. Mas, B.; Cladera, A.; Bestard, J.; Muntaner, D.; López, C.E.; Piña, S.; Prades, J. Concrete with Mixed Recycled Aggregates: Influence of the Type of Cement. *Constr. Build. Mater.* **2012**, *34*, 430–441. [CrossRef]
- 65. Qi, B.; Gao, J.; Chen, F.; Shen, D. Evaluation of the Damage Process of Recycled Aggregate Concrete under Sulfate Attack and Wetting-Drying Cycles. *Constr. Build. Mater.* **2017**, *138*, 254–262. [CrossRef]
- 66. Rattanachu, P.; Tangchirapat, W.; Jaturapitakkul, C. Water Permeability and Sulfate Resistance of Eco-Friendly High-Strength Concrete Composed of Ground Bagasse Ash and Recycled Concrete Aggregate. J. Mater. Civ. Eng. 2019, 31, 04019093. [CrossRef]
- 67. Santillán, L.R.; Locati, F.; Villagrán-Zaccardi, Y.A.; Zega, C.J. Long-Term Sulfate Attack on Recycled Aggregate Concrete Immersed in Sodium Sulfate Solution for 10 Years. *Mater. Constr.* **2020**, *70*, 1–14. [CrossRef]

- 68. Al-Baghdadi, H.M. Experimental Study on Sulfate Resistance of Concrete with Recycled Aggregate Modified with Polyvinyl Alcohol (PVA). *Case Stud. Constr. Mater.* 2021, 14, e00527. [CrossRef]
- 69. Santillán, L.R.; Villagrán Zaccardi, Y.A.; Zega, C.J.; Méndez, E.M. Macroscopic Behavior and Microstructural Analysis of Recycled Aggregate Mortar Bars Exposed to External Sulfate Attack. *Cem. Concr. Compos.* **2023**, *143*, 105277. [CrossRef]
- Shayan, A.; Xu, A. Performance and Properties of Structural Concrete Made with Recycled Concrete Aggregate. ACI Mater. J. 2003, 100, 371–380. [CrossRef]
- 71. Somna, R.; Jaturapitakkul, C.; Amde, A.M. Effect of Ground Fly Ash and Ground Bagasse Ash on the Durability of Recycled Aggregate Concrete. *Cem. Concr. Compos.* **2012**, *34*, 848–854. [CrossRef]
- 72. Tangchirapat, W.; Khamklai, S.; Jaturapitakkul, C. Use of Ground Palm Oil Fuel Ash to Improve Strength, Sulfate Resistance, and Water Permeability of Concrete Containing High Amount of Recycled Concrete Aggregates. *Mater. Des.* **2012**, *41*, 150–157. [CrossRef]
- 73. Tovar-rodríguez, G.; Barra, M.; Pialarissi, S.; Aponte, D.; Vázquez, E. Expansion of Mortars with Gypsum Contaminated Fine Recycled Aggregates. *Constr. Build. Mater.* **2013**, *38*, 1211–1220. [CrossRef]
- Xiao, Q.H.; Li, Q.; Cao, Z.Y.; Tian, W.Y. The Deterioration Law of Recycled Concrete under the Combined Effects of Freeze-Thaw and Sulfate Attack. *Constr. Build. Mater.* 2019, 200, 344–355. [CrossRef]
- 75. Xie, F.; Li, J.; Zhao, G.; Wang, C.; Wang, Y.; Zhou, P. Experimental Investigations on the Durability and Degradation Mechanism of Cast-in-Situ Recycled Aggregate Concrete under Chemical Sulfate Attack. *Constr. Build. Mater.* **2021**, 297, 123771. [CrossRef]
- 76. Xie, F.; Li, J.; Zhao, G.; Zhou, P.; Zheng, H. Experimental Study on Performance of Cast-in-Situ Recycled Aggregate Concrete under Different Sulfate Attack Exposures. *Constr. Build. Mater.* **2020**, *253*, 119144. [CrossRef]
- 77. Xu, F.; Wang, S.; Li, T.; Liu, B.; Li, B.; Zhou, Y. The Mechanical Properties of Tailing Recycled Aggregate Concrete and Its Resistance to the Coupled Deterioration of Sulfate Attack and Wetting–Drying Cycles. *Structures* **2020**, *27*, 2208–2216. [CrossRef]
- Xu, F.; Wang, S.; Li, T.; Liu, B.; Zhao, N.; Liu, K. The Mechanical Properties and Resistance against the Coupled Deterioration of Sulfate Attack and Freeze-Thaw Cycles of Tailing Recycled Aggregate Concrete. *Constr. Build. Mater.* 2021, 269, 121273. [CrossRef]
- 79. Arafa, M.; Tayeh, B.A.; Alqedra, M.; Shihada, S.; Hanoona, H. Investigating the Effect of Sulfate Attack on Compressive Strength of Recycled Aggregate Concrete. J. Eng. Res. Technol. JERT 2017, 4, 137–143.
- Yammine, A.; Hamdadou, M.; Leklou, N.; Bignonnet, F.; Choinska-colombel, M. Effect of Recycled Concrete Aggregates and Recycled Filler on Delayed Ettringite Formation: An Experimental Study Compared to Chemical Modelling. *Cem. Concr. Compos.* 2022, 132, 104636. [CrossRef]
- 81. Yammine, A.; Leklou, N.; Choinska, M.; Bignonnet, F.; Mechling, J. DEF Damage in Heat Cured Mortars Made of Recycled Concrete Sand Aggregate. *Constr. Build. Mater.* **2020**, *252*, 119059. [CrossRef]
- 82. Zajac, M.; Skibsted, J.; Lothenbach, B.; Bullerjahn, F.; Skocek, J.; Ben Haha, M. Effect of Sulfate on CO2 Binding Efficiency of Recycled Alkaline Materials. *Cem. Concr. Res.* 2022, 157, 106804. [CrossRef]
- 83. Zega, C.J.; Dos, G.S.C.; Villagrán-zaccardi, Y.A.; Maio, A.A. Di Performance of Recycled Concretes Exposed to Sulphate Soil for 10 Years. *Constr. Build. Mater.* **2016**, 102, 714–721. [CrossRef]
- 84. Zhang, H.; Ji, T.; Liu, H. Performance Evolution of Recycled Aggregate Concrete (RAC) Exposed to External Sulfate Attacks under Full-Soaking and Dry-Wet Cycling Conditions. *Constr. Build. Mater.* **2020**, 248, 118675. [CrossRef]
- 85. Zhang, H.; Liu, W.; Zhang, J.; Liu, F.; Lin, X.; Ji, T. A New Look at the Resistance of Recycled Aggregate Concrete (RAC) to the External Sulfate Attacks: The Influence of the Multiple Mesoscopic Material Phases. J. Build. Eng. 2023, 64, 105653. [CrossRef]
- Zhao, Z.; Wang, S.; Lu, L.; Gong, C. Evaluation of Pre-Coated Recycled Aggregate for Concrete and Mortar. *Constr. Build. Mater.* 2013, 43, 191–196. [CrossRef]
- Santillan, L.R.; Villagrán Zaccardi, Y.A.; Zega, C.J. External Sulphate Attack on Recycled Concrete: Assessment of the Influence of Recycled Aggregate. *RILEM Bookser.* 2020, 21, 45–51. [CrossRef]
- 88. Li, H.; Liu, J.; Chu, F. Study on Mix Proportion Design Based on Strength and Sulfate Resistance of 100% Recycled Aggregate Concrete. *Buildings* **2022**, *12*, 1467. [CrossRef]
- 89. Zhang, H.; Ji, T.; Liu, H. Performance Evolution of the Interfacial Transition Zone (ITZ) in Recycled Aggregate Concrete under External Sulfate Attacks and Dry-Wet Cycling. *Constr. Build. Mater.* **2019**, *229*, 116938. [CrossRef]
- Arredondo Rea, S.P.; Corral Higuera, R.; Almaral Sánchez, J.L.; Castorena González, J.H.; Neri Flores, M.; Martínez Villafañe, A.; Almeraya Calderón, F. Efficiency of Supplementary Materials Against Steel Corrosion in Concrete with Recycled Aggregate Exposed to Sulfates. ECS Trans. 2009, 20, 499–506. [CrossRef]
- 91. Boudali, S.; Kerdal, D.E.; Ayed, K.; Abdulsalam, B.; Soliman, A.M. Performance of Self-Compacting Concrete Incorporating Recycled Concrete Fines and Aggregate Exposed to Sulphate Attack. *Constr. Build. Mater.* **2016**, *124*, 705–713. [CrossRef]
- 92. Brekailo, F.; Pereira, E.; Pereira, E.; Farias, M.M.; Medeiros-Junior, R.A. Red Ceramic and Concrete Waste as Replacement of Portland Cement: Microstructure Aspect of Eco-Mortar in External Sulfate Attack. *Clean. Mater.* **2022**, *3*, 100034. [CrossRef]
- Bulatovic, V.; Melesev, M. Evaluation of Sulfate Resistance of Concrete with Recycled and Natural Aggregates. Constr. Build. Mater. 2017, 152, 614–631. [CrossRef]
- 94. Candamano, S.; Tassone, F.; Iacobini, I.; Crea, F.; De Fazio, P. The Properties and Durability of Self-Leveling and Thixotropic Mortars with Recycled Sand. *Appl. Sci.* **2022**, *12*, 2732. [CrossRef]
- 95. Santhanam, M.; Cohen, M.D.; Olek, J. Mechanism of Sulfate Attack: A Fresh Look. Part 1: Summary of Experimental Results. *Cem. Concr. Res.* 2002, 32, 915–921. [CrossRef]

- 96. Santhanam, M.; Cohen, M.D.; Olek, J. Mechanism of Sulfate Attack: A Fresh Look Part 2. Proposed Mechanisms. *Cem. Concr. Res.* 2003, *33*, 341–346. [CrossRef]
- 97. Neville, A. The Confused World of Sulfate Attack on Concrete. Cem. Concr. Res. 2004, 34, 1275–1296. [CrossRef]
- 98. Ikumi, T.; Cavalaro, S.H.P.; Segura, I. The Role of Porosity in External Sulphate Attack. *Cem. Concr. Compos.* **2019**, *97*, 1–12. [CrossRef]
- 99. Feng, P.; Garboczi, E.J.; Miao, C.; Bullard, J.W. Microstructural Origins of Cement Paste Degradation by External Sulfate Attack. *Constr. Build. Mater.* **2015**, *96*, 391–403. [CrossRef]
- 100. Kunther, W.; Lothenbach, B.; Scrivener, K.L. On the Relevance of Volume Increase for the Length Changes of Mortar Bars in Sulfate Solutions. *Cem. Concr. Res.* 2013, *46*, 23–29. [CrossRef]
- Irassar, E.F. Sulfate Attack on Cementitious Materials Containing Limestone Filler—A Review. Cem. Concr. Res. 2009, 39, 241–254.
   [CrossRef]
- 102. Rossetti, A.; Falcone, D.; Irassar, E.F. Expansión de Cemento Portland con Diferentes Porcentajes de Filler Calcáreo Frente al Ataque sulfato a distintas temperaturas. In Proceedings of the VII Congreso Internacional-21<sup>a</sup> Reunión Técnica de la AATH, Salta, Argentina, 28–30 September 2016; pp. 437–444.
- Bonen, D.; Cohen, M.D. Magnesium Sulfate Attack on Portland Cement Paste-I. Microstructural Analysis. Cem. Concr. Res. 1992, 22, 169–180. [CrossRef]
- Gollop, R.S.; Taylor, H.F.W. Microstructural and Microanalytical Studies of Sulfate Attack III. Sulfate-Resisting Portland Cement: Reactions with Sodium and Magnesium Sulfate Solutions. Cem. Concr. Res. 1995, 25, 1581–1590. [CrossRef]
- 105. Rodriguez-Navarro, C.; Doehne, E.; Sebastian, E. How Does Sodium Sulfate Crystallize? Implications for the Decay and Testing of Building Materials. *Cem. Concr. Res.* 2000, *30*, 1527–1534. [CrossRef]
- 106. Loser, R.; Leemann, A. An Accelerated Sulfate Resistance Test for Concrete. Mater. Struct. Constr. 2016, 49, 3445–3457. [CrossRef]
- 107. Collepardi, M. Ettringite Formation and Sulfate Attack on Concrete. ACI Symp. Publ. 2001, 200, 21–38. [CrossRef]
- Alexander, M.; Bertron, A.; De Belie, N. Performance of Cement-Based Materials in Aggressive Aqueous Environments; Alexander, M., Bertron, A., De Belie, N., Eds.; Springer: Berlin/Heidelberg, Germany, 2013; ISBN 9789400754126.
- 109. Collepardi, M. A State-of-the-Art Review on Delayed Ettringite Attack on Concrete. *Cem. Concr. Compos.* 2003, 25, 401–407. [CrossRef]
- Monteiro, P.J.M.; Kurtis, K.E. Time to Failure for Concrete Exposed to Severe Sulfate Attack. Cem. Concr. Res. 2003, 33, 987–993.
   [CrossRef]
- 111. Irassar, E.F.; González, M.; Rahhal, V. Sulphate Resistance of Type V Cements with Limestone Filler and Natural Pozzolana. *Cem. Concr. Compos.* 2000, 22, 361–368. [CrossRef]
- Ferraris, C.F.; Stutzman, P.E.; Peltz, M.A.; Winpigler, J.A. Developing a More Rapid Test to Assess Sulfate Resistance of Hydraulic Cements. J. Res. Natl. Inst. Stand. Technol. 2005, 110, 529–540. [CrossRef] [PubMed]
- Messad, S.; Carcasses, M.; Linger, L. Design of an Accelerated Test Method for External Sulfate Attack. In Proceedings of the RILEM Workshop on Long-Term Performance of Cementitious Barriers and Reinforced Concrete in Nuclear Power Plants and Waste Management, Cadarache, France, 30 March–2 April 2009.
- 114. ASTM C 1012; Standard Test Method for Length Change of Hydraulic-Cement Mortars Exposed to a Sulfate Solution. ASTM: West Conshohocken, PA, USA, 2012.
- 115. Suma, M.F.; Santhanam, M. Influence of the Specimen Size on the Expansion of Portland Cement Mortar Immersed in Sodium Sulphate Solution. In *External Sulfate Attack—Field Aspects and Lab Tests*; Menéndez-Mendez, E., Baroghel-Bouny, V., Eds.; Springer: Berlin/Heidelberg, Germany, 2020; p. 163.
- 116. Hossack, A.M.; Thomas, M.D.A. The Effect of Temperature on the Rate of Sulfate Attack of Portland Cement Blended Mortars in Na<sub>2</sub>SO<sub>4</sub> Solution. *Cem. Concr. Res.* **2015**, *73*, 136–142. [CrossRef]
- 117. Wang, K.; Guo, J.; Yang, L.; Zhang, P.; Xu, H. Multiphysical Damage Characteristics of Concrete Exposed to External Sulfate Attack: Elucidating Effect of Drying–Wetting Cycles. *Constr. Build. Mater.* **2022**, *329*, 127143. [CrossRef]
- 118. Brunetaud, X.; Khelifa, M.R.; Al-Mukhtar, M. Size Effect of Concrete Samples on the Kinetics of External Sulfate Attack. *Cem. Concr. Compos.* **2012**, *34*, 370–376. [CrossRef]
- El-Hachem, R.; Rozire, E.; Grondin, F.; Loukili, A. New Procedure to Investigate External Sulphate Attack on Cementitious Materials. *Cem. Concr. Compos.* 2012, 34, 357–364. [CrossRef]
- 120. Girardi, F.; Vaona, W.; Di Maggio, R. Resistance of Different Types of Concretes to Cyclic Sulfuric Acid and Sodium Sulfate Attack. *Cem. Concr. Compos.* **2010**, *32*, 595–602. [CrossRef]
- Yin, Y.; Hu, S.; Lian, J.; Liu, R. Fracture Properties of Concrete Exposed to Different Sulfate Solutions under Drying-Wetting Cycles. Eng. Fract. Mech. 2022, 266, 108406. [CrossRef]
- 122. Gómez-Soberón, J.M. Relationship between Gas Adsorption and the Shrinkage and Creep of Recycled Aggregate Concrete. *Cem. Concr. Aggreg.* 2003, *2*, 42–48.
- Levy, S.M.; Helene, P. Durability of Recycled Aggregates Concrete: A Safe Way to Sustainable Development. *Cem. Concr. Res.* 2004, 34, 1975–1980. [CrossRef]
- 124. Tam, V.W.Y.; Butera, A.; Le, K.N.; Li, W. Utilising CO 2 Technologies for Recycled Aggregate Concrete: A Critical Review. *Constr. Build. Mater.* **2020**, 250, 118903. [CrossRef]

- 125. Omary, S.; Ghorbel, E.; Wardeh, G. Relationships between Recycled Concrete Aggregates Characteristics and Recycled Aggregates Concretes Properties. *Constr. Build. Mater.* **2016**, *108*, 163–174. [CrossRef]
- 126. Thomas, C.; Setién, J.; Polanco, J.A.; Cimentada, A.I.; Medina, C. Influence of Curing Conditions on Recycled Aggregate Concrete. *Constr. Build. Mater.* **2018**, 172, 618–625. [CrossRef]
- 127. Boudali, S.; Soliman, A.M.; Abdulsalam, B.; Ayed, K.; Kerdal, D.E.; Poncet, S.; Materials, A. Microstructural Properties of the Interfacial Transition Zone and Strength Development of Concrete Incorporating Recycled Concrete Aggregate. *Int. J. Struct. Constr. Eng.* 2017, *11*, 1012–1016.
- 128. Poon, C.S.; Shui, Z.H.; Lam, L. Effect of Microstructure of ITZ on Compressive Strength of Concrete Prepared with Recycled Aggregates. *Constr. Build. Mater.* 2004, *18*, 461–468. [CrossRef]
- 129. Etxeberria, M.; Vázquez, E.; Marí, A. Microstructure Analysis of Hardened Recycled Aggregate Concrete. *Mag. Concr. Res.* 2006, 58, 683–690. [CrossRef]
- 130. Otsuki, N.; Miyazato, S.; Yodsudjai, W. Influence of Recycled Aggregate on Interfacial Transition Zone, Strength, Chloride Penetration and Carbonation of Concrete. *J. Mater. Civ. Eng.* **2003**, *15*, 443–451. [CrossRef]
- 131. Corinaldesi, V.; Moriconi, G. Influence of Mineral Additions on the Performance of 100% Recycled Aggregate Concrete. *Constr. Build. Mater.* **2009**, *23*, 2869–2876. [CrossRef]
- 132. Belin, P.; Habert, G.; Thiery, M.; Roussel, N. Cement Paste Content and Water Absorption of Recycled Concrete Coarse Aggregates. *Mater. Struct. Constr.* **2014**, *47*, 1451–1465. [CrossRef]
- 133. Evangelista, L.; Guedes, M.; De Brito, J.; Ferro, A.C.; Pereira, M.F. Physical, Chemical and Mineralogical Properties of Fine Recycled Aggregates Made from Concrete Waste. *Constr. Build. Mater.* **2015**, *86*, 178–188. [CrossRef]
- 134. Sánchez De Juan, M.; Alaejos Gutiérrez, P. Influence of Attached Mortar Content on the Properties of Recycled Concrete Aggregate. In Proceedings of the International RILEM Conference "The Use of Recycled Materials in Building and Structures", Barcelona, Spain, 9–11 November 2004; pp. 536–544.
- 135. Sánchez de Juan, M.; Alaejos Gutiérrez, P. Study on the Influence of Attached Mortar Content on the Properties of Recycled Concrete Aggregate. *Constr. Build. Mater.* **2009**, *23*, 872–877. [CrossRef]
- 136. Etxeberria, M.; Vázquez, E.; Marí, A.; Barra, M. Influence of Amount of Recycled Coarse Aggregates and Production Process on Properties of Recycled Aggregate Concrete. *Cem. Concr. Res.* **2007**, *37*, 735–742. [CrossRef]
- 137. Sosa, M.E.; Villagrán Zaccardi, Y.A.; Zega, C.J. A Critical Review of the Resulting Effective Water-to-Cement Ratio of Fine Recycled Aggregate Concrete. *Constr. Build. Mater.* **2021**, *313*, 125536. [CrossRef]
- Zhao, Y.; Zeng, W.; Zhang, H. Properties of Recycled Aggregate Concrete with Different Water Control Methods. Constr. Build. Mater. 2017, 152, 539–546. [CrossRef]
- 139. Marchi, T.; García-Díaz, E.; Salgues, M.; Souche, J.C.; Devillers, P. Internal Curing Capacity of Recycled Coarse Aggregates Incorporated in Concretes with Low Water/Cement Ratios. *Constr. Build. Mater.* **2023**, 409, 133893. [CrossRef]
- Koga, H.; Katahira, H.; Shimata, A. The Introduction of Recycled-Aggregate Concrete Specifications in Japan and the Research into the Freezing–Thawing Resistance of Recycled-Aggregate Concrete. J. Mater. Cycles Waste Manag. 2022, 24, 1207–1215. [CrossRef]
- 141. Malolepszy, J.; Grabowska, E. Sulphate Attack Resistance of Cement with Zeolite Additive. *Procedia Eng.* **2015**, *108*, 170–176. [CrossRef]
- 142. Naik, N.N.; Jupe, A.C.; Stock, S.R.; Wilkinson, A.P.; Lee, P.L.; Kurtis, K.E. Sulfate Attack Monitored by MicroCT and EDXRD: Influence of Cement Type, Water-to-Cement Ratio, and Aggregate. *Cem. Concr. Res.* **2006**, *36*, 144–159. [CrossRef]
- 143. Müllauer, W.; Beddoe, R.E.; Heinz, D. Sulfate Attack Expansion Mechanisms. Cem. Concr. Res. 2013, 52, 208–215. [CrossRef]
- 144. Fonseca, N.; De Brito, J.; Evangelista, L. The Influence of Curing Conditions on the Mechanical Performance of Concrete Made with Recycled Concrete Waste. *Cem. Concr. Compos.* **2011**, *33*, 637–643. [CrossRef]
- 145. Sosa, M.E.; Zega, C.J.; DI Maio, A.A. Understanding the Influence of Properties of Fine Recycled Aggregates on Recycled Concrete. *Proc. Inst. Civ. Eng. Constr. Mater.* **2021**, 176, 150–160. [CrossRef]
- Berndt, M.L. Properties of Sustainable Concrete Containing Fly Ash, Slag and Recycled Concrete Aggregate. *Constr. Build. Mater.* 2009, 23, 2606–2613. [CrossRef]
- 147. Corral-Higuera, R.; Arredondo-Rea, S.P.; Neri-Flores, M.; Gomez-Soberón, J.M.; Almaral-Sánchez, J.L.; Castorena-González, J.H.; Martínez-Villafañe, A.; Almeraya-Calderon, F. Chloride Ion Penetrability and Corrosion Behavior of Steel in Concrete with Sustainability Characteristics. *Int. J. Electrochem. Sci.* 2011, 6, 958–970. [CrossRef]
- Li, W.; Xiao, J.; Sun, Z.; Kawashima, S.; Shah, S.P. Interfacial Transition Zones in Recycled Aggregate Concrete with Different Mixing Approaches. Constr. Build. Mater. 2012, 35, 1045–1055. [CrossRef]
- Ann, K.Y.; Moon, H.Y.; Kim, Y.B.; Ryou, J. Durability of Recycled Aggregate Concrete Using Pozzolanic Materials. *Waste Manag.* 2008, 28, 993–999. [CrossRef] [PubMed]
- 150. Tam, V.W.Y.; Gao, X.F.; Tam, C.M. Microstructural Analysis of Recycled Aggregate Concrete Produced from Two-Stage Mixing Approach. *Cem. Concr. Res.* 2005, *35*, 1195–1203. [CrossRef]
- 151. Bui, N.K.; Satomi, T.; Takahashi, H. Improvement of Mechanical Properties of Recycled Aggregate Concrete Basing on a New Combination Method between Recycled Aggregate and Natural Aggregate. *Constr. Build. Mater.* 2017, 148, 376–385. [CrossRef]
- 152. Kim, H.; Park, S.; Kim, H. The Optimum Production Method for Quality Improvement of Recycled Aggregates Using Sulfuric Acid and the Abrasion Method. *Int. J. Environ. Res. Public Health* **2016**, *13*, 769. [CrossRef] [PubMed]

- Shaban, W.M.; Yang, J.; Su, H.; Mo, K.H.; Li, L.; Xie, J. Quality Improvement Techniques for Recycled Concrete Aggregate: A Review. J. Adv. Concr. Technol. 2019, 17, 151–167. [CrossRef]
- 154. Zhang, H.; Ji, T.; Liu, H.; Su, S. Improving the Sulfate Resistance of Recycled Aggregate Concrete (RAC) by Using Surface-Treated Aggregate with Sulfoaluminate Cement (SAC). *Constr. Build. Mater.* **2021**, 297, 123535. [CrossRef]
- Nedeljković, M.; Visser, J.; Šavija, B.; Valcke, S.; Schlangen, E. Use of Fine Recycled Concrete Aggregates in Concrete: A Critical Review. J. Build. Eng. 2021, 38, 102196. [CrossRef]
- 156. Sánchez De Juan, M.; Alaejos Gutiérrez, P. Influence of Recycled Aggregate Quality on Concrete Properties. In Proceedings of the International RILEM Conference "The Use of Recycled Materials in Building and Structures", Barcelona, Spain, 9–11 November 2004; pp. 545–553.
- Evangelista, L.; de Brito, J. State-of-the-Art on the Use of Fine Recycled Aggregates in Concrete Production. In Proceedings of the 2nd International RILEM Conference on Progress of Recycling in the Built Environment, Sao Paulo, Brazil, 2–4 December 2009; pp. 175–183.
- Leemann, A.; Loser, R. Accelerated Sulfate Resistance Test for Concrete—Chemical and Microstructural Aspects. In Proceedings of the Second International Conference on Microstructural-Related Durability of Cementitious Composites, Amsterdam, The Netherlands, 11–13 April 2012; pp. 1–12.
- 159. Menéndez, E.; García-Rovés, R.; Aldea, B.; Ruíz, S.; Baroghel-Bouny, V. Combination of Immersion and Semi-Immersion Tests to Evaluate Concretes Manufactured with Sulfate-Resisting Cements. J. Sustain. Cem. Mater. 2019, 8, 337–352. [CrossRef]
- 160. Ikumi, T.; Segura, I. Numerical Assessment of External Sulfate Attack in Concrete Structures. A Review. *Cem. Concr. Res.* 2019, 121, 91–105. [CrossRef]
- 161. Lothenbach, B.; Bary, B.; Le Bescop, P.; Schmidt, T.; Leterrier, N. Sulfate Ingress in Portland Cement. *Cem. Concr. Res.* 2010, 40, 1211–1225. [CrossRef]
- 162. Hodhod, O.A.; Salama, G. Simulation of Expansion in Cement Based Materials Subjected to External Sulfate Attack. *Ain Shams Eng. J.* **2014**, *5*, 7–15. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.